Author’s response to reviews


Authors:

Dirk E Bock (dirk.bock@lhsc.on.ca)
Tracy Robinson (Tracy.Robinson@lhsc.on.ca)
Jamie A Seabrook (Jamie.Seabrook@lhsc.on.ca)
Meghan Rombeek (Meghan.Rombeek@lhsc.on.ca)
Kambiz Norozi (Kambiz.Norozi@lhsc.on.ca)
Guido Filler (Guido.Filler@lhsc.on.ca)
Ralf Rauch (rrauch@khrmk.de)
Cheril L Clarson (Cheril.Clarson@lhsc.on.ca)

Version: 4 Date: 11 September 2014

Author’s response to reviews:

Please see attached Response Letter, thank you.